

Reviewer #1: Microscopes are helpful when making precise incisions in order to protect surrounding tissues. The use of microscope-assisted transduodenal local excision in ampullary tumors was successfully implemented and very well presented in this report.

*Many thanks for your encouraging comments. We believe this paper will be helpful for the readers.*